Author's response to reviews

Title: Detecting depressive and anxiety disorders in distressed patients in primary care; comparative diagnostic accuracy of the Four-Dimensional Symptom Questionnaire (4DSQ) and the Hospital Anxiety and Depression Scale (HADS)

Authors:

Berend Terluin (b.terluin@vumc.nl)
Evelien PM Brouwers (e.p.m.brouwers@uvt.nl)
Harm WJ van Marwijk (hwj.vanmarwijk@vumc.nl)
Peter FM Verhaak (p.verhaak@nivel.nl)
Henriëtte E van der Horst (he.vanderhorst@vumc.nl)

Version: 3 Date: 8 August 2009

Author's response to reviews:

I have effectuated the requested changes:
- The vertical lines have been removed from the tables.
- The coloured text has been removed.
- The titles and legends of the tables have been adapted.
- I have run a final check for typographical and grammatical errors, and made a few corrections.